Author's response to reviews

Title: Neovascular Glaucoma: A Retrospective Review From a Tertiary Center in China

Authors:

Na Liao (nana_jane@163.com)
Chao Hong Li (aoeight@hotmail.com)
Hui Lv Jiang (hljiang314@163.com)
Ai Wu Fang (953736995@qq.com)
Sheng Jie Zhou (425760739@qq.com)
Qin Mei Wang (wqm3@mail.eye.ac.cn)

Version: 5 Date: 23 November 2015

Author's response to reviews:

1. Authors' contribution
   HLJ involved in data collection.
   revised: HLJ was involved in data collection.

2. Authorship
   I have informed all authors and they would send you an email to confirm that.

3. Figure legend
   Censored data related to the ones who had not observed failure at that time
   (including missing data).
   revised:
   Censored data relates to the ones who had not observed failure at that time
   (including missing data).